Author's response to reviews

Title: Unilateral thalamic infarction presenting as vertical gaze palsy

Authors:

Muhib Khan (mkhan4@hfhs.org)
Christos Sidiropoulos (chris.sid@gmail.com)
Panayiotis Mitsias (pmitsias@gmail.com)

Version: 2 Date: 4 May 2011

Author's response to reviews: see over
Dear Editor,

Thanks a lot for reviewing our article and providing us with valuable input to make this case report worthy of publication. We have addressed the issues raised by the reviewer’s as listed below:

1. Vertebral artery dissection was ruled out by cerebral angiogram and etiology was determined to be small vessel disease after extensive workup which include transesophageal cardiogram to look for cardioembolic source.
2. Abstract wording has been changed as recommended with more focus on clinico-radiological correlation.
3. Newer studies are referenced.
4. T-2 and T-2 Flair images of the thalamic infarct are provided.

I hope that you will find the article worthy of publication

Thankyou again.

Muhib Khan MD
Senior Neurology Resident
Henry Ford Health Systems, Detroit, MI
USA